Author's response to reviews

Title: Baropodometry on women suffering from chronic pelvic pain - a cross-sectional study

Authors:

   Carolina Kaercher (ckaercher@hotmail.com)
   Vanessa K Genro (vanessagenro@gmail.com)
   Carlos A Souza (souza.cab@gmail.com)
   Mariane Alfonsin (marianealfonsin@ibest.com.br)
   Greice Berton (g.berton@terra.com.br)
   Joao S Cunha Filho (sabino@vi-rs.net)

Version: 4 Date: 16 November 2011

Author's response to reviews: see over
November 16, 2011.

*Prof. Jigisha Patel,*
*Editor-in-Chief*
*BMC Womens Health*

Dear Prof,

We would like to thank you and the referees for the careful reading and suggestions to our paper entitled “Baropodometry On Women Suffering From Chronic Pelvic Pain – a cross sectional study” MS: 8621039795681698. We performed the final revision of our text and we hope that the final reviewing process finds the manuscript acceptable for publication in the journal.

Sincerely

Carlos Souza,
Hospital de Clinicas de Porto Alegre
Serviço de Ginecologia e Obstetrícia
Rua Ramiro Barcellos, 2350- 11° andar
Porto Alegre – RGS – Brazil
FAX: +55-51-33300034
Email: souza.cab@gmail.com